Author’s response to reviews

Title: Productive activities, mental health and quality of life in disability: Exploring the role enhancement and the role strain hypotheses

Authors:

Christine Fekete (christine.fekete@paraplegie.ch)
Johannes Siegrist (johannes.siegrist@med.uni-duesseldorf.de)
Marcel Post (m.post@dehoogstraat.nl)
Martin Brinkhof (martin.brinkhof@paraplegie.ch)

Version: 4 Date: 14 Dec 2018

Author’s response to reviews:

WE have now included a statement at the end of the Abstract section detailing that the abstract had been presented as a POSTER at the European Congress of Epidemiology and published as a special issue in the journal "Revue d'Epidemiologie et de Santé Publique", Volume 66, Supplement 5.